Author's response to reviews

Title: Energy and Macronutrient Intake and Dietary Pattern among School Children in Bahrain: a cross-sectional study

Authors:
Nadia Gharib (gharibn@gmail.com)
Parveen Rasheed (parveenrasheed1@yahoo.com)

Version: 10 Date: 29 May 2011

Author's response to reviews:

Key Words: Energy and macronutrient intake, food frequency, energy % fat, saturated fat, overweight and obesity, Bahraini school children.